Author’s response to reviews

Title: Graphical comparisons of relative disease burden across multiple risk factors

Authors:

John Ferguson (john.ferguson@nuigalway.ie)

Neil O'Leary (neil.oleary@nuigalway.ie)

Fabrizio Maturo (fabrizio.maturo@nuigalway.ie)

Salim Yusuf (yusufs@mcmaster.ca)

Martin O'Donnell (martin.odonnell@nuigalway.ie)

Version: 3 Date: 16 Aug 2019

Author’s response to reviews:

1. Please include an abstract after your title page. Your abstract should include the following headings: Background, Methods, Results, Conclusions, and should be no longer than 350 words. Please also include a keywords section, which should three to ten keywords.

Background: the context and purpose of the study
Menthod: how the study was performed and statistical tests used
Results: the main findings
Conclusions: brief summary and potential implications

Please ensure you apply this format to the abstract in the main text and the submission system also.

Author response: Thank you. For some reason, the Conclusions section of the abstract was missing. We have added this into the revised paper

2. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Author response: We have supplemented the funding section with this information

3. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures,
tables and additional/supplementary files are cited within the text.

We have included clean versions of all of the Figures and Supplementary material as separate files.